Author's response to reviews

Title: Deficiencies in the transfer and availability of clinical trials evidence: a survey of existing systems and standards

Authors:

Gert van Valkenhoef (g.h.m.van.valkenhoef@rug.nl)
Tommi Tervonen (tervonen@ese.eur.nl)
Bert de Brock (e.o.de.brock@rug.nl)
Hans Hillege (h.hillege@umcg.nl)

Version: 2 Date: 5 November 2011

Author's response to reviews:

Dear editor,

Please consider our revised manuscript, "Deficiencies in the transfer and availability of clinical trials evidence: a survey of existing systems and standards" for publication in BMC Medical Informatics and Decision Making.

According to your wishes, we have revised and expanded the methods section based on the PRISMA statement. We agree that this greatly enhances the usefulness of the paper. For your convenience, a per-item table has been attached using the "confidential comments to the editors" form.

Kindest regards,

Gert van Valkenhoef